

Citation: Barrett C, Chiphwanya J, Mkwanda S, Matipula DE, Ndhlovu P, Chaponda L, et al. (2024) The national distribution of lymphatic filariasis cases in Malawi using patient mapping and geostatistical modelling. PLoS Negl Trop Dis 18(3): e0012056. https://doi.org/10.1371/journal. pntd.0012056

Editor: Alberto Novaes Ramos, Jr, Federal University of Ceará, Fortaleza, Brazil, BRAZIL

Received: September 29, 2023

Accepted: March 10, 2024

Published: March 25, 2024

Copyright: © 2024 Barrett et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its supporting information files.

Funding: This study was supported by the Centre for Neglected Tropical Diseases, Liverpool School of Tropical Medicine, through funding from the Department for International Development (DFID) and GlaxoSmithKline (GSK) awarded to LK-H & MJT for programmatic support for the elimination of lymphatic filariasis as a public health problem. RESEARCH ARTICLE

The national distribution of lymphatic filariasis cases in Malawi using patient mapping and geostatistical modelling

Carrie Barrett^{1*}, John Chiphwanya², Square Mkwanda², Dorothy E. Matipula², Paul Ndhlovu², Limbikani Chaponda², Joseph D. Turner¹, Emanuele Giorgi³, Hannah Betts¹, Sarah Martindale¹, Mark J. Taylor¹, Jonathan M. Read^{3®}, Louise A. Kelly-Hope^{1,4®}

1 Centre for Neglected Tropical Disease, Department of Tropical Disease Biology, Liverpool School of Tropical Medicine, Pembroke Place, Liverpool, United Kingdom, 2 National Lymphatic Filariasis Elimination Programme, Ministry of Health, Lilongwe, Malawi, 3 Lancaster Medical School, South West Drive, Bailrigg, Lancaster, United Kingdom, 4 Department of Livestock and One Health, Institute of Infection, Veterinary and Ecological Sciences, University of Liverpool, Liverpool, United Kingdom

These authors contributed equally to this work.

Abstract

Background

In 2020 the World Health Organization (WHO) declared that Malawi had successfully eliminated lymphatic filariasis (LF) as a public health problem. Understanding clinical case distributions at a national and sub-national level is important, so essential care packages can be provided to individuals living with LF symptoms. This study aimed to develop a national database and map of LF clinical cases across Malawi using geostatistical modelling approaches, programme-identified clinical cases, antigenaemia prevalence and climate information.

Methodology

LF clinical cases identified through programme house-to-house surveys across 90 sub-district administrative boundaries (Traditional Authority (TA)) and antigenaemia prevalence from 57 sampled villages in Malawi were used in a two-step geostatistical modelling process to predict LF clinical cases across all TAs of the country. First, we modelled antigenaemia prevalence in relation to climate covariates to predict nationwide antigenaemia prevalence. Second, we modelled clinical cases for unmapped TAs based on our antigenaemia prevalence spatial estimates.

Principle findings

The models estimated 20,938 (95% Crl 18,091 to 24,071) clinical cases in unmapped TAs (70.3%) in addition to the 8,856 (29.7%), programme-identified cases in mapped TAs. In total, the overall national number of LF clinical cases was estimated to be 29,794 (95% Crl 26,957 to 32,927). The antigenaemia prevalence and clinical case mapping and modelling

^{*} Carrie.Barrett@lstmed.ac.uk

This work was supported by a Medical Research Council Doctoral Training Partnership for CB [project number 2267306]. The funders of this study had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

found the highest burden of disease in Chikwawa and Nsanje districts in the Southern Region and Karonga district in the Northern Region of the country.

Conclusions

The models presented in this study have facilitated the development of the first national LF clinical case database and map in Malawi, the first endemic country in sub-Saharan Africa. It highlights the value of using existing LF antigenaemia prevalence and clinical case data together with modelling approaches to produce estimates that may be used for the WHO dossier requirements, to help target limited resources and implement long-term health strategies.

Author summary

Lymphatic filariasis (LF) is a disfiguring and painful Neglected Tropical Disease, transmitted by mosquitoes, and impairs affected individual's mental wellbeing, social participation, and ability to work. The two most common clinical manifestations are hydrocoele (scrotal swelling) and lymphoedema (swelling of the limbs). Estimates of LF clinical case numbers are required to provide national and local care needs assessment, and for elimination and surveillance purposes. Clinical case prevalence is currently not readily available or is unknown across many sub-Saharan African countries, however Malawi is unique as the LF Programme has conducted extensive house-to-house patient mapping activities across one third of the country. We used this clinical data in combination with measurements of LF infection prevalence and high-resolution climate information, to develop geostatistical models, which estimate the number of clinical cases in unmapped areas. This led to the development of a national database and map of clinical case estimates that will help the Malawi LF Elimination Programme to optimize limited resources, target morbidity management and disability prevention, and improve quality of life of the people affected by this disabling and disfiguring disease.

Introduction

In sub-Saharan Africa lymphatic filariasis (LF) is a mosquito-borne disease caused by the parasitic nematode, *Wuchereria bancrofti* [1,2]. LF is targeted for elimination in 27 African countries (77%) by 2030 described in the World Health Organization (WHO) Neglected Tropical Disease (NTD) road map 2021–2030 [3,4]. To achieve validation of elimination of LF as a public health problem granted by the WHO, countries are required to submit and meet WHO dossier requirements [5]. In addition, countries must be implementing post-validation activities for surveillance and integrating morbidity management into existing health systems [5–7]. In 2018, approximately 51 million individuals were estimated to be infected with LF, which has reduced by 74% since the Global Programme to Eliminate LF (GPELF) began [8]. In 2000, clinical case estimates in sub-Saharan Africa ranged from 46 to 51 million, which are now outdated and clinical case estimates are unavailable in many countries [9]. The two most common chronic clinical manifestations of LF are hydrocoele (scrotal swelling) and lymphoedema (swelling of the limbs) that cause pain, profound disfigurement and large financial, social and mental health losses [2,10].

The Malawi LF Elimination Programme has achieved certification of LF elimination as a public health problem in 2020 from the WHO [3,11]. Over the past two decades the programme successfully implemented prevalence mapping, effective anti-filarial mass drug administration (MDA), impact assessments, morbidity management and disability prevention (MMDP) and operational research activities as outlined in Chiphwanya et al. [11]. The widespread endemic nature of LF across the country became evident in the early 2000s when LF antigenaemia prevalence surveys were conducted in villages using antigen-based immunochromatographic rapid tests (ICTs), although this data was geographically sparse in comparison to large district areas [12–14]. More recently, attention has focussed on obtaining better estimates of clinical burden, and the programme conducted a series of large-scale house-to-house mapping activities, across 23 districts in 90 sub-district administrative areas known as Traditional Authorities (TAs) between 2014–2021. The extensive clinical case mapping covered an area of over 33,000km² populated by 5.6 million people, representing approximately 35% of the geographical area of Malawi, identifying 8,856 clinical cases: male hydrocoele = 6,333 (71.5%; average age 50.5); male lymphoedema = 854 (9.6%; average age 54.4; female lymphoedema = 1,585 (17.9%; average age 50.5); male both = 84 (1.0%); average age 58.8) [11].

In Malawi, estimating the number of hydrocoele and lymphoedema cases was important to allow for the planning and provision of services that are available within the primary care system in all areas with known affected people, in line with the WHO essential package of care recommendations which include: hydrocoele surgeries; treatment for episodes of adenolymphangitis (ADL) through antibiotic treatment and symptomatic management; management of lymphoedema (trained health workers able to provide and teach patients self-care measures of hygiene, skin and wound-care, elevation, and exercise) [15]. In addition, short term studies have shown the inclusion of diaphragmatic deep-breathing exercises, lymphatic massage, and dietary changes to improve lymphoedema status and significantly reduce frequency and duration of ADLs [16–18]. The case estimates also helped to direct actions and provide documentation for the WHO dossier on (1) the number of hydrocoele and lymphoedema case estimates in all endemic implementation units (IUs); (2) assess the availability and quality of available funding and resources to (3) provide full geographical coverage of essential package of care within all endemic IUs [5].

In many endemic countries clinical case estimates are lacking, which is likely due to the significant time, human, and financial resources required [10]. Only one country in Asia has developed a national database and map of LF cases, and also found that disease prevalence was positively correlated with antigenaemia prevalence (prior to MDA) [6], which supports findings from historical studies [19,20]. The close relationship between disease and antigenaemia prevalence suggests that these data may be used together to model and predict clinical cases and prevalence rates in unmapped TAs. The use of multiple types of prevalence data has proven useful when data is sparse and resources limited, as shown in other disease mapping activities [21]. In addition, the use of climate covariates such as temperature, rainfall, humidity, may also help model predictions as found in antigenaemia and microfilaria prevalence studies as they impact the abundance of mosquito vectors and LF transmission rates, therefore may be useful in understanding disease distributions and risk [22–25].

To support the Malawi LF Elimination Programme with obtaining LF case estimates across all endemic areas of the country, this study aimed to develop a national map of LF clinical cases, using a geostatistical modelling approach from a combination of clinical case data, antigenaemia prevalence and climate information. A two-step geostatistical analysis was conducted. First, we modelled antigenaemia prevalence (prior to MDA) in relation to climate covariates to predict national antigenaemia prevalence. Second, we modelled clinical cases for unmapped TAs based on these antigenaemia prevalence spatial estimates.

Methods

Ethics statement

Ethical approval was obtained from the Research Ethics Committee at Liverpool School of Tropical Medicine, UK (protocol number 12.22) and the National Health Sciences Research Committee, Ministry of Health, Malawi (protocol number 1260).

Data sources

LF clinical case data. LF clinical case data was collected across 90 TA areas approximately 35% of Malawi, between 2014 and 2017. National health surveillance assistants, trained to identify and report individual clinical cases (lymphoedema and hydrocoele) using an innovative phone reporting tool [11,26,27]. Data were reported by health facility catchments, which were generally aligned with TA administrative boundaries defined by Humanitarian Data Exchange [28]. Two instances where TA and health facility catchment boundaries were not continuous were resolved by merging TA boundaries for analysis purposes.

LF antigenaemia data. The LF antigenaemia prevalence data was collected from three surveys conducted in selected 57 villages across all districts in Malawi between 2000 and 2003, prior to initiation of MDA [12,13,29]. The ICT diagnostic tool was used to detect the presence of *W. bancrofti*-specific circulating filarial antigen in whole-blood samples. Details on the surveys' methodology are available in the original studies [12,13,29]. 21 TAs with available antigenaemia data and LF clinical case data was summarised in S2 Table.

Statistical analysis

Analytical overview. Exploratory analysis identified antigenaemia as a predictor of clinical case prevalence in TAs where both observations were available, see <u>S1 File</u>. However, antigenaemia data was not available for all areas of Malawi (only 57 villages). Step 1: A geostatistical model was fitted to available antigenaemia prevalence data which incorporated climate covariates to predict and map antigenaemia prevalence in areas with and without available antigenaemia prevalence data. Step 2: A second geostatistical model using the predicted antigenaemia prevalence from step 1 was used to predict LF clinical case estimates.

Predicting antigenaemia prevalence-geostatistical model 1. To predict antigenaemia prevalence in areas with and without data available data, we used available antigenaemia data and climate covariates identified from the literature and based on the assumption of their impact on mosquito vector abundance and LF transmission rates [24,25,30-32]. We fitted a geostatistical model of antigenaemia prevalence, P(x), where x_i is the geolocation of village *i*. Available antigenaemia prevalence was determined by the number of positives, Y, divided by the number of individuals tested, N, following a binomial distribution. Climate covariates: average annual temperature (°C); average annual humidity (kPa); elevation (m); and average annual rainfall (mm) between 1970-2000 were obtained from World Clim, a database for 1km spatial resolution climate surfaces for global land areas. [33]. A Principle Component Analysis (PCA) was performed on the climate covariates to mitigate for collinearity – the non-independence of predictor covariates [34]. The first component of the PCA of the chosen climate covariates for each village location, d(x), was the explanatory variable. To explain the spatial variation in antigenaemia prevalence we include an unobserved stochastic process, S(x), to represent the variation in P(x) that is not explained by d(x). Finally, random variation U assuming a Normal Distribution was included in the model, which gave the following step one geostatistical equation:

$$\log\left(\frac{P(x_i)}{1 - P(x_i)}\right) = d(x_i) + S(x_i) + U_i$$
(Eq1)

Predicting LF clinical cases-geostatistical model 2.

$$M(x_{i}^{*}) = a(x_{i}^{*})^{\tau}\beta + \log(Pop(x_{i}^{*})) + S(x_{i}^{*}) + U_{i}$$
(Eq2)

To predict LF clinical cases in the TA with no data (i.e., unmapped), we fitted a second geostatistical model of LF clinical cases $M(x^*)$ using available clinical case data, where x_j^* is the geolocation for each TA, *j*, assuming a Poisson distribution. Note x^* in geostatistical model 2 has a different spatial scale to *x* in geostatistical model 1. The explanatory variable, $a(x^*)$, predicted antigenaemia prevalence, from geostatistical model 1 was fitted to each centroid geolocation of TA, x^* with available clinical case data, including an offset term of the population size (*Pop*) for each TA, taken from 2018 census mapping [35]. An unobserved stochastic process, $S(x^*)$ was included in the model to represent the variation in $M(x^*)$ that is not explained by $a(x^*)$. Finally, random variation *U* assuming a Normal distribution was included in the model to give the step two geostatistical equation.

All geostatistical analyses were performed in R programming software using the PrevMap R package [36,37]. To assess the goodness of fit of the geostatistical model 2 predictions, the mean predicted LF clinical cases were compared to programme-collected clinical cases numbers within TAs observations [11]. Within S2 File, the goodness of fit of the geostatistical model 1 predictions were assessed by comparison against antigenaemia prevalence.

Mapping

Three maps were produced, the first showing predicted antigenaemia prevalence (%) from the geostatistical model 1 using QGIS software 3.22.5 [38] on 4.5km (longitude) x 4.6km (latitude) scale. The second map showing LF clinical cases and the third showing LF clinical case prevalence per 100,000 population [35] for TA areas [28] from geostatistical model 2 predictions and available programme-collected data.

National estimates

National and TA boundary LF clinical case estimates and prevalence (case per 100,000 population) were summarised from geostatistical model predictions for the 320 unmapped TA areas. <u>S1 Table</u> summarises this for each TA boundary in Malawi. The 95% credible intervals for national LF clinical case estimates were calculated from geostatistical model predictions and summarised for predicted case numbers. National sex-specific hydrocoele and lymphoedema clinical cases were estimated by partitioning the national estimated LF clinical cases by programme-identified observed proportions.

Results

Antigenaemia prevalence mapping-geostatistical model 1

We found a positive association between antigenaemia prevalence and clinical case prevalence within TAs, see S1 File. The antigenaemia prevalence predicted from the first geostatistical model using climate covariates was mapped across Malawi and is presented in Fig 1. LF antigenaemia prevalence varied between 0.5% and 69.3% across the country on a spatial scale of 4.5km x 4.6km cells. For the majority of the country the prevalence was low (0.9% - 15.7%),



Fig 1. LF prevalence (antigenaemia) map from step one geostatistical analysis. Note: Maps were produced in QGIS mapping software (https://qgis.org) using the base layer from OpenStreetMap (https://www.openstreetmap.org/), and country administrative boundaries available from the Humanitarian Data Exchange [28].

https://doi.org/10.1371/journal.pntd.0012056.g001

however the prevalence was high (1.7% - 50.5%) in the northern district Karonga, districts along the southern shore of Lake Malawi (Salima, Dedza, Ntcheu, Mangochi, Balaka and Machinga) (1.4% - 33.0%), and highest in the southern districts of Chikwawa and Nsanje (16.4% - 69.3%).

Clinical case mapping-geostatistical model 2

The number of predicted LF clinical cases for each TA across Malawi from the second geostatistical model was presented in Fig 2A. The predicted LF clinical case prevalence per 100,000 population for each TA is presented in Fig 2B. The number of cases within each TA were summarised in S1 Table.

The number of LF clinical cases varied between TAs, particularly within the middle region of the country, in areas along the southern shore of Lake Malawi and programme-collected TA data, however the prevalence of cases is more consistent amongst predicted TAs in the middle region of the country, suggesting differing population size does impact the number of cases found within each TA. Two TAs with the highest number of LF clinical cases in the middle region of the country were in Lilongwe district, called TA Mazengera and TA Kabudula. Similar to the antigenaemia prevalence, LF clinical cases and case prevalence were high in the northern Karonga district, areas along the southern shore of Lake Malawi, and highest in the southern districts of Chikwawa and Nsanje.

Note: Maps were produced in QGIS mapping software (<u>https://qgis.org</u>) using the base layer from OpenStreetMap (<u>https://www.openstreetmap.org</u>/), and country administrative boundaries available from the Humanitarian Data Exchange [28].

Assessing the goodness of fit

To assess the goodness of fit of the geostatistical model in step 2, the predicted prevalence LF clinical cases was compared to the number of programme identified clinical case prevalence per 100,000 population shown in Fig 3. Many TA observations that fell on, or very close to, the line, suggesting that the geostatistical model has a good predictive ability.



Fig 2. National LF clinical case (A) numbers and (B) prevalence (cases per 100,000 population) and (C) population size taken from Census Report in 2018 [35] summarised at Traditional Authority level.

https://doi.org/10.1371/journal.pntd.0012056.g002



Fig 3. Predicted clinical case prevalence from geostatistical analysis compared against programme identified LF clinical case prevalence per 100,000 population.

https://doi.org/10.1371/journal.pntd.0012056.g003

National LF clinical cases

From geostatistical modelling, the overall estimated number of LF clinical cases was 29,794 calculated with 95% credible intervals (CrI) 26,957 to 32,927 across all endemic districts in Malawi. The geostatistical analysis identified a further estimated 20,938 (95% CrI 18,091 to 24,071) cases in the 320 unmapped TA areas (70.3%), in addition to the 8,856 cases identified by the programme house-to-house surveys (29.7%), see Table 1. We estimated that there were

	Clinical cases	95% Credible Interval	Total Population*	Clinical Case Prevalence (cases per 100,000 population)
Mapped TA Areas	8,856	NA	5,613,230	158
Unmapped TA Areas	20,938	18,091-24,071	11,950,519	175
All TA Areas	29,794	26,957-32,927	17,563,749	170

* Population size taken from Census Report in 2018 [35].

https://doi.org/10.1371/journal.pntd.0012056.t001

21,306 (71.5%) male hydrocoele cases, 2,873 (9.6%) male lymphoedema, 5,332 (17.9%) female lymphoedema and 283 (1.0%) male both.

Discussion

This study makes Malawi the first LF endemic African country to produce a national level database of LF clinical cases, estimating 29,794 (26,957–32,927) from the 8,856 (29.7%) programme-identified cases and 20,938 (18,091–24,071) cases predicted from geostatistical analysis. As well as producing a set of risk maps of LF clinical case and antigenaemia prevalence estimates. The highest numbers of LF clinical cases and antigenaemia prevalence from geostatistical model predictions and programme data were observed in Chikwawa and Nsanje districts in the Southern Region and Karonga district in the Northern Region of the country. The results presented in this study provide the Malawi LF programme and health system with an informative understanding of the clinical case distributions across local regions, allowing them to target resources for MMDP in identified high risk TA areas, where cases were predicted as high as 524 (Thyolo district, TA Nchilamwela).

The maps produced in this study demonstrate the widespread distribution of LF clinical cases and prevalence across Malawi. The detailed antigenaemia prevalence map was produced from data collected between 2000–2003, prior to MDA, using climate information from 1970–2000. Following implementation of MDA initiated in 2008, antigenaemia prevalence has decreased [39]. The map shows the highest LF case prevalence occurred in the northern region in Karonga district and southern region, Nsanje and Chikwawa districts, which may be due to these areas having optimum climate conditions that drive transmission and abundance of mosquito vectors. Within these southern regions of Malawi, Nsanje and Chikwawa districts, predominantely *Anopheles funestus*, as well as *A. arabiensis* and *A. gambiae* have been categorised as main LF vectors [11,40]. Along Lake Malawi shore, antigenaemia prevalence was observed to be higher compared to in-land areas, thus likely due to mosquito breeding sites driving *W. bancrofti* infection as well as human populations inhabiting areas close to water bodies [41].

From LF clinical case predictions and programmatic data presented in this study, the highest number of cases occurred within districts with highest antigenaemia prevalence, Chikwawa, Nsanje and Karonga. However, middle regions of Malawi showed a higher prevalence of cases where antigenaemia prevalence was found to be low, suggesting that population may play an important role in defining risk areas for clinical cases. In Bangladesh similar gender ratios and age distributions have been observed [6]. Higher proportions of LF clinical case prevalence been found to be associated with people living in rural areas, poverty, and poor sanitation [42]. Similar comparisons can be drawn to highly endemic districts in Malawi, Chikwawa and Nsanje, although more research is needed to identify risk factors for LF cases within this context. <u>S1 Table</u> describes the case predictions made from this study analysis predictions and programme-identified cases for each TA boundary in Malawi. Our findings suggest that historic antigenaemia prevalence may be a good predictor for highly endemic regions, but more research is needed from other countries and ecological zones to solidify this relationship [6,19,20].

Our geostatistical approach offers an alternative to national patient searching. This approach may be an improvement over other methods, i.e. community drug distributor estimations of cases during MDA programmes, which potentially underestimate clinical cases [43,44]. In the absence of expertise and resources for more complex geostatistical modelling, we advocate that antigenaemia prevalence data could be used to estimate LF clinical case distributions. Other country elimination programmes may refer to the S1 File, which features the

relationship between antigenemia prevalence and LF clinical cases, and to use this as a guide to estimate clinical cases within their implementation unit boundary if antigenemia prevalence is available. Geostatistical modelling has shown to substantially outperform current WHO guidelines to collect case estimates, in studies of other NTDs, including soil-transmitted helminth infections in Kenya, Sierra Leone, and Zimbabwe [45]. Geostatistical modelling approaches can offer improved precision of cases estimates for a reduced field-sampling effort, particularly use-ful for NTDs where resources are often limited. More work utilising geostatistical approaches to estimate and map clinical cases in sub-Saharan Africa would be beneficial as current estimates are limited and many countries have antigenaemia data that could be used [4].

For men with hydrocoeles, we estimate thousands of men living with symptoms are likely to require surgery in the next decade. Assessing the capacity of hospitals, the infrastructure, human and financial resources required for these surgeries is important to determine whether they can be provided routinely through the health system. A hospital facility assessment to determine the readiness and quality of hydrocelectomy services has been conduced in Malawi in 2019 [46]. To reduce the backlog of cases, hydrocoele campaigns were conduced periodically from 2008 completing more than 1500 surgeries in high burden districts, Karonga, Chikwawa and Nsanje [11]. In Malawi, hydrocoele surgeries were found to have significant improvements in men's quality of life, as well as life-time economic benefits to the individual, his family, and his community which greatly outweigh the low cost of surgery (estimated at US \$68 during campaigns) [47,48]. However, since 2015 when the LF programme hydrocoele campaigns have ceased, the capacity of hospital's to address the backlog of hydrocoeles remains a challenge in TA's where LF clinical cases are high.

For people with lymphoedema, life-long home-based MMDP strategies is required to manage their and ADLs, to prevent and hinder progression of their symptoms [15]. Studies have shown that lymphoedema predominately affects women [26,49–51], although the reasons for this are not well understood [52]. The home-based care recommended by WHO includes daily hygiene, skin care, limb exercise and elevation [15], with recent research showing the benefits of additional exercises including lymphatic massage, deep diaphragm breathing techniques, skin mobilization, seated and standing exercises [16–18]. Most mild stages of lymphoedema will manage symptoms with home-based care activities, however the more severe stages require more specialist care [50]. Adopting a holistic approach which addresses the physical, psychological and social implications of lymphoedema is required due to the chronic, stigmatizing nature of this condition [3].

In Malawi, lymphatic management training has been provided to all health surveillance assistants and two staff 259 health centres. More than 4000 community health workers have been trained in lymphoedema management and to provide further training in the lymphoedema home-based care to all persons affected identified within their catchment area. All health centres across the country provide free of charge services during ADL episodes, such as antibiotics or pain relief [11]. The findings from this study show the wide spread distribution LF clinical cases across the country, that will all require lymphoedema management training to hinder the progression of their disease and reduce further disability. The majority of LF clinical cases in Malawi are located within rural communities, where access to healthcare or trained healthcare staff remains a challenge, as well as equitable access to quality healthcare services for women [53]. Additional challenges include addressing the psychosocial consequences of lymphoedema [51].

As many countries move into the LF elimination phase, strategies to focus on alleviating suffering of affected individuals through morbidity management is critical. The Malawi LF programme is in the process of planning and continuing the scale-up of a home-based enhanced self-care in all endemic districts, however more funding is required to continue this

vital work. The informative LF case numbers presented in this study will allow the programme to identity where they need to verify the numbers and plan to allocate resources, including training of health staff and health surveillance assistants. This will help the integration of the essential packages of care into health systems for increased sustainability. In addition, it is a WHO post-elimination requirement to estimate LF case numbers at the implementation unit level, as well as conducting post-elimination surveillance [5].

Limitations and considerations

There are several limitations to this study; we will discuss those related to study analysis first. As there was limited data on antigenaemia and LF clinical case mapping, we did not conduct a validation of our geostatistical prediction. This lack of validation means we could not test the reliability of our predictions, therefore more research needs to be done to validate some of the predicted areas through ground truthing field surveys. Additionally, estimates for sex specific hydrocoele and lymphoedema cases were based on the assumption that there were no differences in gender ratios in different TA areas of Malawi and that the programme collected data distributions of hydrocoele and lymphoedema were the same for our model predictions.

Limitations from a public health perspective include that most endemic country programmes have limited access to geostatistical modelers, which restrict the scale up of such work, however a pool of key experts may be convened and work with national programmes to obtain better estimates. Finally, the generalizability in this study is limited to countries with available clinical case data across endemic regions.

Conclusion

Malawi is the first LF endemic African country to produce a national level database and a set of risk maps of LF clinical case and antigenaemia prevalence estimates. This has been achieved by the extensive detailed patient mapping conducted by the Malawi LF programme, which was used in combination with LF prevalence data, climate information and geostatistical analytical methods. This study highlights the value of combining different data in resource limited settings to help to save time, human and financial resources. Additionally, these predictions equip the Malawi national programme and Ministry of Health with information to help assess the readiness and quality of services needed to act on and deliver targeted care to the people who need it most. As well as providing lessons for many of country programmes close to eliminating LF as a public health problem.

Supporting information

S1 STROBE Checklist. Strobe statement. (DOC)

S1 Table. Predicted LF clinical cases and prevalence for Traditional Authority Boundary Administrative Units in Malawi from predictions from geostatistical analysis or programme-collected data. (XLSX)

S2 Table. Traditional Authorities with available antigenaemia prevalence data and LF clinical cases data in Malawi. (DOCX) S1 File. Additional Methods and Results for Exploratory Analysis: Antigenaemia and Clinical Case Prevalence in Malawi.

(DOCX)

S2 File. Assessing goodness of fit of geostatistical model 1. (DOCX)

Acknowledgments

The authors wish to thank the health surveillance assistants, programme coordinators and managers, district health workers, district and regional officials, community leaders and participants who made this study possible.

Author Contributions

Conceptualization: Carrie Barrett, Louise A. Kelly-Hope.

Data curation: John Chiphwanya, Square Mkwanda, Sarah Martindale.

Formal analysis: Carrie Barrett, Jonathan M. Read.

Funding acquisition: John Chiphwanya, Square Mkwanda, Mark J. Taylor, Louise A. Kelly-Hope.

Investigation: Dorothy E. Matipula, Paul Ndhlovu, Limbikani Chaponda, Sarah Martindale.

Methodology: Carrie Barrett, Emanuele Giorgi, Hannah Betts, Sarah Martindale, Jonathan M. Read, Louise A. Kelly-Hope.

Project administration: John Chiphwanya, Square Mkwanda, Sarah Martindale, Louise A. Kelly-Hope.

Resources: John Chiphwanya, Square Mkwanda, Mark J. Taylor, Louise A. Kelly-Hope.

Supervision: John Chiphwanya, Square Mkwanda, Joseph D. Turner, Louise A. Kelly-Hope.

Visualization: Carrie Barrett.

Writing - original draft: Carrie Barrett.

Writing – review & editing: Carrie Barrett, John Chiphwanya, Square Mkwanda, Paul Ndhlovu, Limbikani Chaponda, Joseph D. Turner, Emanuele Giorgi, Hannah Betts, Mark J. Taylor, Jonathan M. Read, Louise A. Kelly-Hope.

References

- World Health Organization. Weekly Epidemiological Record, 2021, vol. 96, 41 [full issue]. Weekly Epidemiological Record. 2021; 96(41):497–508.
- World Health Organization. Lymphatic filariasis 2023 [updated 01/06/2023]. Available from: https://www.who.int/news-room/fact-sheets/detail/lymphatic-filariasis.
- 3. World Health Organization. Ending the neglect to attain the sustainable development goals: a road map for neglected tropical diseases 2021–2030. World Health Organization, 2020.
- World Health Organization. The Expanded Special Project for Elimination of Neglected Tropical Diseases (ESPEN) 2017 Annual Report. Brazzaville: World Health Organization. Regional Office for Africa, 2018 2018. Report No.
- 5. World Health Organization. Validation of elimination of lymphatic filariasis as a public health problem. Geneva: World Health Organization; 2017.
- 6. Karim MJ, Haq R, Mableson HE, Sultan Mahmood ASM, Rahman M, Chowdhury SM, et al. Developing the first national database and map of lymphatic filariasis clinical cases in Bangladesh: Another step

closer to the elimination goals. PLoS Negl Trop Dis. 2019; 13(7):e0007542. Epub 2019/07/16. https:// doi.org/10.1371/journal.pntd.0007542 PMID: 31306409; PubMed Central PMCID: PMC6658114.

- Malecela MN, Mwingira U, Mwakitalu ME, Kabali C, Michael E, Mackenzie CD. The sharp end—experiences from the Tanzanian programme for the elimination of lymphatic filariasis: notes from the end of the road. Annals of Tropical Medicine & Parasitology. 2009; 103(sup1):53–7. <u>https://doi.org/10.1179/</u> 000349809X12502035776676 PMID: 19843398
- Cromwell EA, Schmidt CA, Kwong KT, Pigott DM, Mupfasoni D, Biswas G, et al. The global distribution of lymphatic filariasis, 2000–18: a geospatial analysis. The Lancet Global Health. 2020; 8(9):e1186– e94. https://doi.org/10.1016/S2214-109X(20)30286-2 PMID: 32827480
- Ichimori K, King JD, Engels D, Yajima A, Mikhailov A, Lammie P, et al. Global programme to eliminate lymphatic filariasis: the processes underlying programme success. PLoS Negl Trop Dis. 2014; 8(12): e3328. Epub 20141211. https://doi.org/10.1371/journal.pntd.0003328 PMID: 25502758; PubMed Central PMCID: PMC4263400.
- World Health Organization. Lymphatic filariasis: managing morbidity and preventing disability: an aidemémoire for national programme managers. World Health Organization, 2013 924150529X.
- Chiphwanya J, Mkwanda S, Kabuluzi S, Mzilahowa T, Ngwira B, Matipula DE, Chaponda L, Ndhlova P, Katchika P, Mahebere Chirambo C, Moses P, Kumala J, Chiumia M, Barrett C, Betts H, Fahy J, Rebollo Polo M, Reimer L, Stanton MC, Thomas B, Freer S, Molyneux DH, Bockarie MJ, Mackenzie CD, Taylor MJ, Martindale S, Kelly-Hope LA. Elimination of lymphatic filariasis as a public health problem in Malawi. PLoS Negl Trop Dis. 2024 Feb 16; 18(2):e0011957. https://doi.org/10.1371/journal.pntd.0011957 PMID: 38363794; PMCID: PMC10903958.
- Ngwira BM, Jabu CH, Kanyongoloka H, Mponda M, Crampin AC, Branson K, et al. Lymphatic filariasis in the Karonga district of northern Malawi: a prevalence survey. Ann Trop Med Parasitol. 2002; 96 (2):137–44. Epub 2002/06/26. https://doi.org/10.1179/0003498302125000411 PMID: 12080974.
- Nielsen NO, Makaula P, Nyakuipa D, Bloch P, Nyasulu Y, Simonsen PE. Lymphatic filariasis in Lower Shire, southern Malawi. Trans R Soc Trop Med Hyg. 2002; 96(2):133–8. Epub 2002/06/12. https://doi. org/10.1016/s0035-9203(02)90279-8 PMID: 12055799.
- Ngwira BM, Tambala P, Perez AM, Bowie C, Molyneux DH. The geographical distribution of lymphatic filariasis infection in Malawi. Filaria journal. 2007; 6:12–. <u>https://doi.org/10.1186/1475-2883-6-12</u> PMID: 18047646.
- 15. World Health Organization. Lymphatic filariasis: managing morbidity and preventing disability: an aidemémoire for national programme managers. 2nd ed ed. Geneva: World Health Organization; 2021 2021.
- Douglass J, Mableson HE, Martindale S, Kelly-Hope LA. An Enhanced Self-Care Protocol for People Affected by Moderate to Severe Lymphedema. Methods Protoc. 2019; 2(3). Epub 20190904. https://doi.org/10.3390/mps2030077 PMID: 31487887; PubMed Central PMCID: PMC6789820.
- Douglass J, Hailekiros F, Martindale S, Mableson H, Seife F, Bishaw T, et al. Addition of Lymphatic Stimulating Self-Care Practices Reduces Acute Attacks among People Affected by Moderate and Severe Lower-Limb Lymphedema in Ethiopia, a Cluster Randomized Controlled Trial. J Clin Med. 2020; 9(12). Epub 20201217. <u>https://doi.org/10.3390/jcm9124077</u> PMID: <u>33348721</u>; PubMed Central PMCID: PMC7766500.
- Douglass J, Mableson H, Martindale S, Jhara ST, Karim MJ, Rahman MM, et al. Effect of an Enhanced Self-Care Protocol on Lymphedema Status among People Affected by Moderate to Severe Lower-Limb Lymphedema in Bangladesh, a Cluster Randomized Controlled Trial. J Clin Med. 2020; 9(8). Epub 2020/08/06. https://doi.org/10.3390/jcm9082444 PMID: 32751676; PubMed Central PMCID: PMC7464742.
- Bundy DA, Grenfell BT, Rajagopalan PK. Immunoepidemiology of lymphatic filariasis: the relationship between infection and disease. Immunol Today. 1991; 12(3):A71–5. <u>https://doi.org/10.1016/S0167-5699(05)80021-0 PMID: 2069681.</u>
- Srividya A, Pani SP, Rajagopalan PK, Bundy DA, Grenfell BT. The dynamics of infection and disease in bancroftian filariasis. Trans R Soc Trop Med Hyg. 1991; 85(2):255–9. <u>https://doi.org/10.1016/0035-</u> 9203(91)90046-2 PMID: 1887487.
- Amoah B, Diggle PJ, Giorgi E. A geostatistical framework for combining spatially referenced disease prevalence data from multiple diagnostics. Biometrics. 2020; 76(1):158–70. https://doi.org/10.1111/ biom.13142 PMID: 31449327
- Hassan AN. Bancroftian filariasis: spatial patterns, environmental correlates and landscape predictors of disease risk. J Egypt Soc Parasitol. 2004; 34(2):501–13. Epub 2004/08/04. PMID: 15287173.
- Lindsay SW, Thomas CJ. Mapping and estimating the population at risk from lymphatic filariasis in Africa. Trans R Soc Trop Med Hyg. 2000; 94(1):37–45. Epub 2000/04/05. <u>https://doi.org/10.1016/s0035-9203(00)90431-0 PMID: 10748895</u>.

- Hassan AN, Dister S, Beck L. Spatial analysis of lymphatic filariasis distribution in the Nile Delta in relation to some environmental variables using geographic information system technology. J Egypt Soc Parasitol. 1998; 28(1):119–31. Epub 1998/06/09. PMID: 9617048.
- Stanton MC, Molyneux DH, Kyelem D, Bougma RW, Koudou BG, Kelly-Hope LA. Baseline drivers of lymphatic filariasis in Burkina Faso. Geospat Health. 2013; 8(1):159–73. Epub 2013/11/22. https://doi. org/10.4081/gh.2013.63 PMID: 24258892.
- 26. Stanton MC, Mkwanda SZ, Debrah AY, Batsa L, Biritwum N-K, Hoerauf A, et al. Developing a community-led SMS reporting tool for the rapid assessment of lymphatic filariasis morbidity burden: case studies from Malawi and Ghana. BMC Infectious Diseases. 2015; 15(1):214. <u>https://doi.org/10.1186/ s12879-015-0946-4</u> PMID: 25981497
- Stanton M, Molineux A, Mackenzie C, Kelly-Hope L. Mobile Technology for Empowering Health Workers in Underserved Communities: New Approaches to Facilitate the Elimination of Neglected Tropical Diseases. JMIR public health and surveillance. 2016; 2(1):e2-e. https://doi.org/10.2196/publichealth. 5064 PMID: 27227155.
- The Humanitarian Data Exchange. Malawi Traditional Authority 2018. 2018 Malawi Traditional Authority (Admin3 Border) dataset shared by National Statistical Office during Cyclone Idai]. Available from: https://data.humdata.org/dataset/2018_malawi_ta_dataset-updated-admin3.
- Ngwira B, Tambala P, Maria P, Bowie C, Molyneux D. The geographical distribution of lymphatic filariasis infection in Malawi. Filaria Journal. 2007; 6. <u>https://doi.org/10.1186/1475-2883-6-12</u> PMID: 18047646
- Eneanya OA, Cano J, Dorigatti I, Anagbogu I, Okoronkwo C, Garske T, et al. Environmental suitability for lymphatic filariasis in Nigeria. Parasit Vectors. 2018; 11(1):513. Epub 2018/09/19. https://doi.org/10. 1186/s13071-018-3097-9 PMID: 30223860; PubMed Central PMCID: PMC6142334.
- Thompson DF, Malone JB, Harb M, Faris R, Huh OK, Buck AA, et al. Bancroftian filariasis distribution and diurnal temperature differences in the southern Nile delta. Emerg Infect Dis. 1996; 2(3):234–5. https://doi.org/10.3201/eid0203.960313 PMID: 8903237.
- Lardeux F, Cheffort J. Ambient temperature effects on the extrinsic incubation period of Wuchereria bancrofti in Aedes polynesiensis: implications for filariasis transmission dynamics and distribution in French Polynesia. Medical and Veterinary Entomology. 2001; 15(2):167–76. https://doi.org/10.1046/j. 0269-283x.2001.00305.x PMID: 11434550
- **33.** Fick SE, Hijmans RJ. WorldClim 2: new 1-km spatial resolution climate surfaces for global land areas. International Journal of Climatology. 2017; 37(12):4302–15. https://doi.org/10.1002/joc.5086
- Dormann CF, Elith J, Bacher S, Buchmann C, Carl G, Carré G, et al. Collinearity: a review of methods to deal with it and a simulation study evaluating their performance. Ecography. 2013; 36(1):27–46. https://doi.org/10.1111/j.1600-0587.2012.07348.x
- Census Report. 2018 Malawi Population Housing Census: Main Report: National Statistical Office; 2018.
- Giorgi E, Diggle PJ. PrevMap: An R Package for Prevalence Mapping. 2017. 2017; 78(8):29. Epub 2017-06-01. https://doi.org/10.18637/jss.v078.i08
- **37.** R Development Core Team. A language and environment for statistical computing. Vienna, Austria: R Foundationfor Statistical Computing; 2023.
- 38. QGIS Development Team. QGIS Geographic Information System. 2023.
- Neglected Tropical Diseases Support Centre. Malawi Eliminates Lymphatic Filariasis & Other NTD News 2020. Available from: <u>https://www.ntdsupport.org/news/malawi-eliminates-lymphatic-filariasis-other-ntd-news</u>.
- 40. Merelo-Lobo AR, McCall PJ, Perez MA, Spiers AA, Mzilahowa T, Ngwira B, et al. Identification of the vectors of lymphatic filariasis in the Lower Shire Valley, southern Malawi. Transactions of The Royal Society of Tropical Medicine and Hygiene. 2003; 97(3):299–301. <u>https://doi.org/10.1016/s0035-9203</u> (03)90149-0 PMID: 15228246
- Aniaku IE, Onyishi GC, Nwosu CG, Urama CC, Akobe NA, Nnawuihe OO, et al. Predisposing Factors to Lymphatic Filariasis among Residents in Igbo-Eze North: An Endemic Area in Nigeria. Iran J Parasitol. 2021; 16(4):663–71. <u>https://doi.org/10.18502/ijpa.v16i4.7879</u> PMID: <u>35082895</u>; PubMed Central PMCID: PMC8710194.
- 42. Williams T, Karim MJ, Uddin S, Jahan S, Asm SM, Forbes SP, et al. Socio-economic and environmental factors associated with high lymphatic filariasis morbidity prevalence distribution in Bangladesh. Plos Neglect Trop Dis. 2023; 17(7):e0011457. https://doi.org/10.1371/journal.pntd.0011457 PMID: 37432968

- Mathieu E, Amann J, Eigege A, Richards F, Sodahlon Y. Collecting baseline information for national morbidity alleviation programs: different methods to estimate lymphatic filariasis morbidity prevalence. The American journal of tropical medicine and hygiene. 2008; 78(1):153–8. PMID: 18187799
- 44. Smith EL, Mkwanda SZ, Martindale S, Kelly-Hope LA, Stanton MC. Lymphatic filariasis morbidity mapping: a comprehensive examination of lymphoedema burden in Chikwawa district, Malawi. Transactions of The Royal Society of Tropical Medicine and Hygiene. 2014; 108(12):751–8. <u>https://doi.org/10.1093/trstmh/tru150 PMID: 25282001</u>
- 45. Johnson O, Fronterre C, Amoah B, Montresor A, Giorgi E, Midzi N, et al. Model-Based Geostatistical Methods Enable Efficient Design and Analysis of Prevalence Surveys for Soil-Transmitted Helminth Infection and Other Neglected Tropical Diseases. Clin Infect Dis. 2021; 72(Suppl 3):S172–s9. https:// doi.org/10.1093/cid/ciab192 PMID: 33905476; PubMed Central PMCID: PMC8201574.
- 46. Martindale S, Mableson H, Bodimeade C, Hume H, Badia X, Karim J, et al. The development and rollout of a new hydrocoele surgery facility assessment tool for the elimination of lymphatic filariasis. Int Health. 2022; 14(Suppl 2):ii55–ii63. https://doi.org/10.1093/inthealth/ihac020 PMID: 36130253; PubMed Central PMCID: PMC9492276.
- Betts H, Martindale S, Chiphwanya J, Mkwanda S, Matipula D, Ndhlovu P, et al. Significant improvement in quality of life following surgery for hydrocoele caused by lymphatic filariasis in Malawi: A prospective cohort study. Plos Neglect Trop Dis. 2020; 14:e0008314. <u>https://doi.org/10.1371/journal.pntd.0008314</u> PMID: 32384094
- Sawers L, Stillwaggon E, Chiphwanya J, Mkwanda SZ, Betts H, Martindale S, et al. Economic benefits and costs of surgery for filarial hydrocele in Malawi. Plos Neglect Trop Dis. 2020; 14(3):e0008003. https://doi.org/10.1371/journal.pntd.0008003 PMID: 32210436
- Mwingira U, Chikawe M, Mandara WL, Mableson HE, Uisso C, Mremi I, et al. Lymphatic filariasis patient identification in a large urban area of Tanzania: An application of a community-led mHealth system. Plos Neglect Trop Dis. 2017; 11(7):e0005748-e. <u>https://doi.org/10.1371/journal.pntd.0005748</u> PMID: 28708825.
- Kebede B, Martindale S, Mengistu B, Kebede B, Mengiste A, H/Kiros F, et al. Integrated morbidity mapping of lymphatic filariasis and podoconiosis cases in 20 co-endemic districts of Ethiopia. Plos Neglect Trop Dis. 2018; 12(7):e0006491. https://doi.org/10.1371/journal.pntd.0006491 PMID: 29965963
- Barrett C, Chiphwanya J, Chaponda L, Matipula DE, Turner J, Taylor MJ, et al. Mental Health Conditions in People Affected by Filarial Lymphoedema: prevalence, associated risk factors and the impact of an enhanced self-care intervention. International Health. 2023.
- Allotey P, Gyapong M. The gender agenda in the control of tropical diseases: A review of current evidence. The Special Topics in Social, Economic and Behavioural (SEB) Research. 2005.
- Shomuyiwa DO, George NS, Sunday BA, Omotayo FO, Mwaba M, David SC, et al. Addressing neglected tropical diseases in Africa: A gender perspective. Health Science Reports. 2023; 6(11): e1726. https://doi.org/10.1002/hsr2.1726 PMID: 38028711